Author's response to reviews

Title: Electrophysiological Differences in Hirayama Disease, Amyotrophic Lateral Sclerosis and Cervical Spondylotic Amyotrophy

Authors:

Xiang Jin (jinxiang81@hotmail.com)
Jianyuan Jiang (jianyuanjiang05@126.com)
Feizhou Lv (lufeizhou@hotmail.com)
Xinlei Xia (dr.xinlei@gmail.com)
Lixun Wang (wanglxun@163.com)
Chaojun Zheng (countd2388@163.com)

Version: 4 Date: 26 September 2014

Author's response to reviews: see over
1. Please remove additional file from the submission system.
Response: It has been removed.

2. Kindly include all email addresses of all authors on the title page. The title page should contain, at minimum, the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.
Response: All email addresses of all authors have been included on the title page.

3. Please add the context of study within background section of the abstract.
Response: Context of study has been added to the background section of the abstract.

4. Kindly use authors initials on the Author's contribution section. Please note that for manuscripts with more than one author, all BMC Series journals require an Authors' Contributions section to be placed after the Competing Interests section.
   An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. To qualify as an author one should 1) have made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; 2) have been involved in drafting the manuscript or revising it critically for important intellectual content; and 3) have given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content. Acquisition of funding, collection of data, or general supervision of the research group, alone, does not justify authorship.
   We suggest the following format (please use initials to refer to each author's contribution): AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript.
   Contributors who do not meet the criteria for authorship should be listed in an acknowledgements section. Examples of those who might be acknowledged include a person who provided purely technical help, writing assistance, or a department chair who provided only general support.
Response: Authors initials have been used on the Author's contribution section. Authors' Contributions section have been placed after the Competing Interests section.